Author’s response to reviews

Title: Prognostic Value of Total Tumor Volume in Patients with Nasopharyngeal Carcinoma Treated with Intensity-Modulated Radiotherapy

Authors:

Shao-Bo Liang (liangshb@sysucc.org.cn)
Jian-Jian Teng (deartjj@163.com)
Xue-Feng Hu (hxfeng@fsyyy.com)
Xing-Li Yang (yangxingl@sysucc.org.cn)
Min Luo (luomin@sysucc.org.cn)
Xiao-Na Fang (fangxn@sysucc.org.cn)
Dong-Sheng Liu (ldsheng@fsyyy.com)
Yong Chen (chenyong@sysucc.org.cn)
Li-Wu Fu (fulw@sysucc.org.cn)

Version: 3 Date: 23 Jun 2017

Author’s response to reviews:

1. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

   Answer:

   Thank you for this comment.

   We have represented authors' names using their full initials instead of their full name in the Authors’ Contributions section. The amendment has been made to the manuscript (Author Contributions section, line 386-391, page 17).

2. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship). Currently, the contributions of "carrying out data collection" do not
automatically qualify them for authorship. Please provide clarification on their contributions, or remove their names from the list of authors and place them in the “Acknowledgements” section instead. Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group - alone - does not usually justify authorship.

Answer:

Thank you for your helpful comment.

SBL and JJT contributed to the conception and design of the study, data acquisition, data analysis, data interpretation and the draft of the manuscript. XFH, XLY, ML and XNF carried out the acquisition and interpretation of data. DSL performed the data analysis. YC and LWF contributed with the conception and design of the study, data acquisition, data interpretation and critical edit of the manuscript. All authors read and approved the final manuscript.

The amendment has been made to the manuscript (Author Contributions section, line 386-391, page 17).

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer:

Thank you for this comment.

We have uploaded the manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files are also clean versions. Figures have been uploaded as separate files.

4. Additionally, according to the requirement of our cancer center, the authenticity of this article has been validated by uploading the key raw data onto the Research Data Deposit public platform (www.researchdata.org.cn), with the approval RDD number as RDDA2017000217. The amendment has been made to the manuscript (Methods section, line 109-112, page 6).